Dear Editor,

Following the decision taken by the Editors and the comments made by the Reviewers, I am sending to you a new version of our paper "A new extract of the plant calendula officinalis produces a dual in vitro effect: cytotoxic anti-tumor activity and lymphocyte activation" (MS: 7988736339546422).

The revised version has the following changes:
- The Background section of Abstract has been modified.
- The abbreviations have been placed after the conclusions.
- The competing interest has been revised.
- The figures have been cropped, and the figure title removed.

Regards,

Angel Miguel Garcia